

Individual Rather Than Household Euler Equations: Identification and Estimation of Individual Preferences Using Household Data

Maurizio Mazzocco

CCPR-059-06

December 2006

California Center for Population Research On-Line Working Paper Series

Individual Rather Than Household Euler Equations: Identification and Estimation of Individual Preferences Using Household Data^{*}

Maurizio Mazzocco[†]

Current Draft, December 2006. First Draft, March 2005.

Abstract

In this paper it is shown that the intratemporal and intertemporal preferences of each decision maker in the household can be identified even if individual consumption is not observed. This identification result is used jointly with the Consumer Expenditure Survey (CEX) to estimate the intratemporal and intertemporal features of individual preferences. This paper is one of the first attempts to provide estimates of the wife's and husband's intertemporal preferences by taking into account that household behavior is the outcome of joint decisions. The empirical findings indicate that there is heterogeneity in intertemporal preferences between wife and husband. The identification and estimation results are important for at least two reasons. First, they suggest that to answer policy questions the household decision process should be characterized using one set of preferences for each decision maker. Second, the estimates of individual preferences provided in this paper can be used to evaluate policies aimed at affecting household intertemporal behavior.

^{*}I thank the National Science Foundation for Grant SES-0231560 and the Graduate School, University of Wisconsin-Madison, for their support. I am very grateful to Orazio Attanasio, Pierre André Chiappori, James Heckman, John Kennan, and Annamaria Lusardi for their insight and suggestions. I would also like to thank the participants at SITE 2004, Econometric Society Summer Meeting 2004, and seminars at CREST-Paris, Duke University, Stanford University, University of California at Berkeley, University of Maryland, University of Minnesota, University of North Carolina, University of Wisconsin-Madison, for their invaluable comments and Lucas Davis for excellent research assistance. Errors are mine.

[†]University of Wisconsin-Madison

1 Introduction

The evaluation of many public policies requires knowledge of the preferences that determine the behavior of multiperson households. Changes in tax rates on pension assets, asset-based means-tested welfare programs, and marriage penalty relief programs are only a few examples. The traditional approach for estimating preferences assumes that households behave as single agents. Under this assumption, each household can be characterized using a unique utility function independently of the household structure. Since the unique utility function depends on household total consumption, which is observed, the intratemporal and intertemporal features of household preferences can be identified and estimated using standard methods.

Numerous papers have rejected the hypothesis that households behave as single agents. For instance, the results of Schultz (1990), Thomas (1990), Browning et al. (1994), Browning and Chiappori (1998), and Mazzocco (2005) indicate that micro-level data are not consistent with this hypothesis. The main implication of this finding is that public policies cannot be evaluated using a unique household utility function, since as shown in Mazzocco (2004) important aspects of intra-household risk sharing and specialization are ignored. Estimates of the preferences of each decision maker in the household are required. The main obstacle in the identification and estimation of individual preferences is that they depend on individual consumption, which is generally not observed. The goal of this paper is to identify and estimate such preferences using the limited amount of information which is available in household surveys. This is one of the first attempts to identify and estimate the intertemporal features of individual preferences by taking into account that household-level data are the outcome of joint decisions by household members.

This paper makes two main contributions. First, it is shown that the preferences of each decision maker in the household can be identified even if individual consumption is not observed, provided that household consumption, individual labor supply, and individual wages are observed. To illustrate the idea behind this result, consider a married couple. If individual consumption were observed, individual preferences could be identified by standard methods using individual Euler equations, i.e. one set of intertemporal optimality conditions for each agent, and intraperiod optimality conditions. Individual consumption is generally not observed, but household consumption, individual labor supply, and wages provide information on this variable. In particular, if at least one agent works in each period, the marginal rate of substitution between individual consumption and leisure should equal the real wage. As a consequence, this agent's consumption can be written as a function of labor supply and wages. Since consumption of the second agent is equal to the difference between total household consumption and consumption of the first agent, the spouse's consumption can also be written as a function of observed variables. These functions can be used to substitute out individual consumption from the marginal utilities that define the individual Euler equations and intraperiod optimality conditions. It can then be shown that these reduced-form optimality conditions and variations in household consumption, individual labor supply, and wages provide sufficient information to recover the original utility functions.

As a second contribution, individual preferences are estimated using the described identification result, a specific functional form for the individual utility functions, and data from the CEX. To evaluate the performance of the identification result, individual preferences are first estimated for single females and males with no children. For this group of households, individual consumption is observed since it is equivalent to household consumption. Individual preferences can therefore be estimated using the identification method proposed in this paper as well as standard methods. The results indicate that the identification result are comparable to the estimates obtained using standard methods. The empirical findings also suggest that there is heterogeneity in intertemporal preferences between single females and single males: the intertemporal elasticity of substitution of single males is more than twice the corresponding elasticity for single females.

The identification result is then applied to a sample of couples. Similarly to single individuals, I find strong evidence of heterogeneity in intertemporal preferences between wives and husbands. In particular, the intertemporal elasticity of substitution of wives is about half the elasticity for husbands or, equivalently in this paper, wives are about twice as risk averse as husbands. A comparison of the parameter estimates for single and married agents indicates that single males are less risk averse than married males and that single females are more risk averse than married females.

These findings have one main implication. In Mazzocco (forthcoming), it is shown that households behave as single agents only if individual preferences belong to the Harmonic Absolute Risk Aversion (HARA) class with identical curvature parameter. The preference heterogeneity found in this paper indicates that this condition is not satisfied. Therefore economists and policy makers should not rely on preference estimates obtained using the standard unitary model to evaluate alternative policy recommendations. Instead, policy analysis should be performed using individual preferences and the corresponding parameter estimates.

This paper is related to the literature on the collective representation of household behavior. Manser and Brown (1980) and McElroy and Horney (1981) are the first papers to characterize the household as a group of agents making joint decisions. In those papers the household decision process is modeled as a Nash bargaining problem. Chiappori (1988; 1992) extends their analysis to allow for any type of efficient decision process. The theoretical model used in the present paper is an intertemporal generalization of Chiappori's collective model.

The intraperiod features of individual preferences have been identified and estimated in other papers. For instance, Blundell et al. (2001), Chiappori (1988; 1982), Chiappori et al. (2004), Donni (2004), Fong and Zhang (2001) show that different aspects of intraperiod preferences can be identified. Donni (2004) also estimates them. The present paper is, however, one of the first attempts

to identify and estimate the intertemporal features of individual preferences using household data.

The paper is organized as follows. In section 2, the individual Euler equations are derived. Section 3 outlines the identification procedure. Section 4 describes the empirical implementation. Section 5 discusses econometric issues. Section 6 describes the data and section 7 presents the estimation results. Section 8 concludes.

2 Household and Individual Euler Equations

Consider a two-person household living for \mathcal{T} periods in an uncertain environment. In each period $t \in \{0, ..., \mathcal{T}\}$ and state of nature $\omega \in \Omega$, member *i* receives non-labor income $y^i(t, \omega)$, consumes a private composite good in quantity $c^i(t, \omega)$ and supplies labor in quantity $h^i(t, \omega)$. Let $C(t, \omega)$ be household total private consumption and let $l^i(t, \omega) = T - h^i(t, \omega)$ be leisure of member *i*, where T is the time available to each spouse in each period. The price of private consumption will be denoted by $p(t, \omega)$ and agent *i*'s wage by $w^i(t, \omega)$. Household members can save jointly by using a risk-free asset. Denote by $s(t, \omega)$ and R(t), respectively, the amount of wealth invested in the risk-free asset and its gross return.¹ Each household member is characterized by individual preferences, which are assumed to be separable over time and across states of nature. The corresponding utility function U_i is assumed to be increasing, concave, and twice continuously differentiable. Agent *i*'s utility function can depend on agent *j*'s private consumption and leisure but only additively, i.e.

$$U^{i}(c^{1}, c^{2}, l^{1}, l^{2}) = u^{i}(c^{i}, l^{i}) + \delta_{i}u^{j}(c^{j}, l^{j})$$

where δ_i is the altruism parameter. Let β_i be the discount factor of member *i*.

The next two subsections describe two different approaches to identifying and estimating the intertemporal and intratemporal features of the preferences that characterize household decisions.

2.1 Household Euler Equations

The theoretical and empirical literature on intertemporal decisions has traditionally assumed that households behave as single agents independently of the number of decision makers. This is equivalent to assuming that the utility functions of the individual members can be collapsed into a unique utility function which fully describes the preferences of the entire household. Following this approach, suppose that household preferences can be represented using a unique von Neumann-Morgenstern utility function $U(C, l^1, l^2)$ and a household discount factor β . Intertemporal decisions

¹The results of the paper are still valid if risky assets are introduced in the model.

can then be determined by solving the following problem:²

$$\max_{\{C_t, l_t^1, l_t^2, s_t\}} E_0 \left[\sum_{t=0}^{\mathcal{T}} \beta^t U\left(C_t, l_t^1, l_t^2\right) \right]$$

$$s.t. \ p_t C_t + s_t \le \sum_{i=1}^2 \left(y_t^i + w_t^i h_t^i \right) + R_t s_{t-1} \quad \forall t, \omega$$

$$s_T \ge 0 \quad \forall \omega.$$

$$(1)$$

The first order conditions of the unitary model (1) can be used to derive the following standard household Euler equations for consumption:

$$U_C\left(C_t, l_t^1, l_t^2\right) = \beta E_t \left[U_C\left(C_{t+1}, l_{t+1}^1, l_{t+1}^2\right) R_{t+1} \frac{p_t}{p_{t+1}} \right]$$

Since the variables defining these intertemporal optimality conditions are observed in various datasets, in the past two decades the standard household Euler equations have been used to test the intertemporal decisions of the household and to estimate the parameters that characterize its behavior.

This approach has one major limitation: the parameter estimates of the intertemporal unitary model can be used to understand household behavior and to answer policy questions only if households behave as single agents. Mazzocco (forthcoming) shows that this assumption is satisfied if and only if the following strong restrictions on individual preferences are satisfied: (i) household members have identical discount factors; (ii) the individual preferences belong to the HARA class and have identical curvature parameters. The evidence based on household Euler equations indicates that this assumption is violated. In particular, in the past twenty years economists have rejected household Euler equations using either the sample of couples or the sample of couples jointly with singles.³ Mazzocco (2005) estimates the standard household Euler equations for couples and separately for singles using the CEX and the Panel Study of Income Dynamics (PSID). He finds that the standard household Euler equations are rejected for couples, but not for singles. Two additional tests based on household Euler equations are performed in Mazzocco (2005; forthcoming) and the outcome suggests that the behavior of a group of agents differs from the behavior of single agents. Additional evidence against the unitary model has been collected in a static framework for instance by Thomas (1990), Browning, Bourguignon, Chiappori and Lechene (1994), Browning and Chiappori (1998).

These empirical findings indicate that it may be important to estimate an alternative model that better characterizes the intertemporal behavior of the household.

²The dependence on the states of nature is suppressed to simplify the notation.

³See Browning and Lusardi (1995) for a survey.

2.2 Individual Euler Equations

This section relaxes the assumption that the individual utility functions can be collapsed into a unique utility function. Without this restriction, it must be established how individual preferences are aggregated to determine household decisions. Following Chiappori (1988; 1992) and Mazzocco (2004; forthcoming), it is assumed that every decision is on the ex-ante Pareto frontier, which implies that household intertemporal behavior can be characterized as the solution of the following Pareto problem:

$$\max_{\{c_t^1, c_t^2, l_t^1, l_t^2, s_t\}} E_0 \left[\sum_{t=0}^{\mathcal{T}} \beta_1^t u^1(c_t^1, l_t^1) \right] + \mu E_0 \left[\sum_{t=0}^{\mathcal{T}} \beta_2^t u^2(c_t^2, l_t^2) \right]$$
(2)
$$s.t. \sum_{i=1}^2 p_t c_t^i + s_t \le \sum_{i=1}^2 \left(y_t^i + w_t^i h_t^i \right) + R_t s_{t-1} \quad \forall t, \omega$$
$$s_T \ge 0 \quad \forall \omega,$$

where μ is a combination of Pareto weights and altruism parameters, and it can be interpreted as the relative decision power at the time of household formation.

Under standard assumptions, the following Euler equations for consumption can be derived:

$$u_{c}^{i}\left(c_{t}^{i}, l_{t}^{i}\right) = \beta_{i} E_{t}\left[u_{c}^{i}\left(c_{t+1}^{i}, l_{t+1}^{i}\right) R_{t+1} \frac{p_{t}}{p_{t+1}}\right] \quad \forall \ i = 1, 2.$$

$$(3)$$

Two remarks are in order. First, the individual Euler equations are not affected by the aggregation problem that affects the standard household Euler equations, since they are satisfied independently of the number of household members. Second, the leisure Euler equations could be added to the consumption Euler equations to characterize the intertemporal behavior of the household. However, they are satisfied only if the corresponding agent supplies a positive amount of labor in each period and state of nature. Since this assumption is excessively strong, only the consumption Euler equations will be employed.

Three main assumptions characterize the intertemporal collective model (2) and the corresponding Euler equations. First, the household Euler equations as well as the individual Euler equations characterize only the intertemporal behavior of households that are not borrowing constrained. There is mixed evidence on the importance of liquidity constraints. For instance, Zeldes (1989) and Gross and Souleles (2002) find that borrowing constraints characterize a significant fraction of the U.S. population. Runkle (1991), Meghir and Weber (1996), and Carneiro and Heckman (2002) find that at most a small fraction of households are liquidity constrained. The theoretical and empirical results of this paper hold for household that are not borrowing constrained in the period considered in the analysis.⁴

⁴Future borrowing constraints affect household decisions in period t and t + 1. This effect is captured in the individual Euler equations by the information set at t. Consequently, as long as the individual Euler equations are satisfied during the survey period, the identification and estimation results hold.

Second, it is assumed that there is no household production or equivalently that household production is determined exogenously. Under this assumption, if individual labor supply is observed, individual leisure is also observed. The generalization of the identification and estimation results to a framework with household production is an important research topic, but it is left for future research.

Third, it is assumed that household decisions are always on the ex-ante Pareto frontier, which implies that the individual members must be able to commit to future allocations of resources at the time of household formation. To test whether the assumption of ex-ante efficiency represents a good approximation of household decisions the following standard efficiency condition will be analyzed jointly with the Euler equations:

$$\frac{u_c^1\left(c_t^1, l_t^1\right)}{u_c^2\left(c_t^2, l_t^2\right)} = \mu.$$
(4)

If individual private consumption and individual labor supply were observed, individual preferences could be estimated using the individual Euler equations and the efficiency condition. Unfortunately, consumption is only measured at the household level. The next section is devoted to showing that the parameters that characterize household intertemporal behavior can be identified using the consumption Euler equations and the intraperiod conditions even if consumption is not observed at the individual level.

3 Identification of Individual Preferences

Consider a household making efficient decisions and suppose that the following variables are observed: household private consumption, individual labor supply, individual wages and interest rate. It is assumed that each agent is characterized by a utility function which is twice continuously differentiable. But the functional form of the utility function is unknown. It will be shown that under these conditions individual preferences can be identified.

Suppose that in each period t at least one agent supplies a positive amount of labor. Without loss of generality, it will be assumed that agent 1 satisfies this restriction. Under this assumption, period t first order conditions of the intertemporal collective model imply that agent 1's marginal rate of substitution between private consumption and leisure must be equal to the real wage, i.e.,

$$\frac{u_l^1\left(c_t^1, T - h_t^1\right)}{u_c^1\left(c_t^1, T - h_t^1\right)} = q\left(c_t^1, h_t^1\right) = \bar{w}_t^1,$$

where $\bar{w}_t^1 = w_t^1 / p_t$.

If the inverse function of q is well-defined, agent 1's consumption can be written as the following unknown function of individual labor supply and real wage:

$$c_t^1 = g\left(\bar{w}_t^1, h_t^1\right),$$

where the parameter T is included in the function g.⁵ Since household private consumption is observed, agent 2's private consumption can also be written as a function of observed variables as follows:

$$c_t^2 = C_t - g\left(\bar{w}_t^1, h_t^1\right).$$

Using the function g, individual private consumption can be substituted out of the marginal utilities that define the Euler equations for private consumption and the intraperiod optimality conditions. Denote with f_k^1 and f_k^2 these transformed marginal utilities with respect to good k for agent 1 and 2. Then f_k^1 and f_k^2 can be defined as follows:

$$f_{k}^{1}\left(\bar{w}_{t}^{1},h_{t}^{1}\right) = u_{k}^{1}\left(g\left(\bar{w}_{t}^{1},h_{t}^{1}\right),T-h_{t}^{1}\right) \quad k = c,l,$$

$$f_{k}^{2}\left(C_{t},\bar{w}_{t}^{1},h_{t}^{1},h_{t}^{2}\right) = u_{k}^{2}\left(C_{t}-g\left(\bar{w}_{t}^{1},h_{t}^{1}\right),T-h_{t}^{2}\right) \quad k = c,l.$$
(5)

The transformed marginal utilities can be used to rewrite the individual private consumption Euler equations as functions of variables that are observed. To that end, the assumption that agent 1 can freely choose and decides to supply a positive amount of labor must be fulfilled for two consecutive periods. Under this restriction, the individual private consumption Euler equations can be written as follows:

$$f_c^1\left(\bar{w}_t^1, h_t^1\right) = \beta_1 E_t \left[f_c^1\left(\bar{w}_{t+1}^1, h_{t+1}^1\right) R_{t+1} \frac{p_t}{p_{t+1}} \right],\tag{6}$$

$$f_c^2\left(C_t, \bar{w}_t^1, h_t^1, h_t^2\right) = \beta_2 E_t \left[f_c^2\left(C_{t+1}, \bar{w}_{t+1}^1, h_{t+1}^1, h_{t+1}^2\right) R_{t+1} \frac{p_t}{p_{t+1}} \right].$$
(7)

Since household private consumption, individual labor supply, individual wages, and the interest rate are observed, the transformed marginal utilities f_c^1 and f_c^2 , and the discount factors β_1 and β_2 can be identified using methods that have been developed for the identification of Euler equations.

The remaining transformed marginal utilities can be identified using the intraperiod optimality conditions. Specifically, agent 1's marginal rate of substitution between private consumption and leisure must be equal to the real wage even if individual consumption is substituted out using the function g. This implies that

$$\frac{u_l^1\left(g\left(\bar{w}_t^1, h_t^1\right), T - h_t^1\right)}{u_c^1\left(g\left(\bar{w}_t^1, h_t^1\right), T - h_t^1\right)} = \frac{f_l^1\left(\bar{w}_t^1, h_t^1\right)}{f_c^1\left(\bar{w}_t^1, h_t^1\right)} = \bar{w}_t^1.$$

Since the function f_c^1 is known for any combination of \bar{w}_t^1 and h_t^1 the transformed marginal utility f_l^1 is identified.

The private consumption efficiency condition can be written using the transformed marginal utilities in the following form:

$$\frac{f_c^1\left(\bar{w}_t^1, h_t^1\right)}{f_c^2\left(C_t, \bar{w}_t^1, h_t^1, h_t^2\right)} = \mu.$$
(8)

⁵The consumption function g is well-defined if the marginal rate of substitution q is strictly increasing in consumption, which is a standard assumption in the labor literature. Lemma 1 in the appendix gives a formal statement of the condition under which g is well-defined. The function g corresponds to the m-consumption function introduced by Browning (1998).

The functions f_c^1 and f_c^2 are known, which implies that the ratio of Pareto weights μ is identified.

Finally, under the additional assumption that agent 2 can choose freely and decides to supply a positive amount of labor, agent 2's transformed marginal utility of leisure can be identified by equating her marginal rate of substitution between private consumption and leisure to the real wage, i.e.,

$$\frac{f_l^2 \left(C_t, \bar{w}_t^1, h_t^1, h_t^2 \right)}{f_c^2 \left(C_t, \bar{w}_t^1, h_t^1, h_t^2 \right)} = \bar{w}_t^2$$

All the transformed marginal utilities are therefore identified. However, the information on individual preferences is contained in the original marginal utilities. The following proposition shows that the original marginal utilities are identified if the transformed marginal utilities are known and variations in household private consumption, individual labor supply, and wages are observed.

Proposition 1 If both agents supply a positive amount of labor and either u^1 or u^2 satisfies the invertibility condition (15), then u_c^1 , u_c^2 , u_l^1 , u_l^2 , μ , and g are identified up to the additive constant of g.

If only agent 1 supplies a positive amount of labor and u^1 satisfies the invertibility condition (15), then u_c^1 , u_c^2 , u_l^1 , μ , and g are identified up to the additive constant of g.

Proof. In the appendix.

To provide the intuition underlying proposition 1, note that by equation (5) for every realization of the exogenous variables agent 2's transformed marginal utility of consumption must satisfy the following equality:

$$f_c^2(C, \bar{w}^1, h^1, h^2) = u_c^2(C - g(\bar{w}^1, h^1), T - h^2).$$
(9)

Consider variations in the exogenous variables that generate a group of households with identical \bar{w}_1 , h_1 , h_2 , but different C. This group of households provides information on $u_{c,c}^2$, i.e. on how agent 2's marginal utility of consumption varies with agent 2's consumption holding everything else constant. To see this observe that f_c^2 is known, which implies that it is known how f_c^2 varies with C if \bar{w}_1 , h_1 , and h_2 are held constant. Since (9) is satisfied for every feasible C, how u_c^2 varies with C holding \bar{w}_1 , h_1 , and h_2 constant must be equivalent to how f_c^2 varies with C if \bar{w}_1 , h_1 , and h_2 constant is equal to $u_{c,c}^2$.

Consider changes in the exogenous variables that generate the group of households for which C, \bar{w}_1 , and h_2 are constant, but h_1 varies. This group of households provides joint information on how agent 2's marginal utility of consumption varies with agent 2's consumption and on how $g(\bar{w}^1, h^1)$ varies with h_1 holding everything else constant. To explain this note that it is known how f_c^2 varies with h_1 if C, \bar{w}_1 , and h_2 are held constant. By (9), how u_c^2 varies with h^1 holding C, \bar{w}_1 , and h_2 constant must be equivalent to how f_c^2 varies with h_1 if C, \bar{w}_1 , and h_2 are held constant. Finally, observe that by varying h^1 on the right hand side of (9) we obtain information on $u_{c,c}^2g_{h_1}$. Consider variations in the exogenous variables that generate the group of households for which C, h_1 , and h_2 are constant, but \bar{w}_1 varies. Using the argument employed for the previous group of households, it can be argued that this set of observations provides information on $u_{c,c}^2 g_{\bar{w}_1}$.

It is therefore possible to identify how $g(\bar{w}^1, h^1)$ varies with h_1 using the first group of households jointly with the second. Using the first and third group of households, it can be identified how $g(\bar{w}^1, h^1)$ varies with \bar{w}_1 . Finally, since it is known how $g(\bar{w}^1, h^1)$ varies with h_1 and with \bar{w}_1 , the function g is also known up to an additive constant. It is then straightforward to recover the original marginal utilities using the transformed marginal utilities and g.

Proposition 1 implies that the individual preferences over private consumption and leisure that determine the intertemporal and intratemporal household decisions can be identified. This leads to the following corollary.

Corollary 1 The individual preferences over private consumption and leisure are identified up to an additive constant.

This section shows that individual preferences can be identified without assuming a particular utility function. In the following sections, specific utility functions will be used jointly with the identification result presented in this section to estimate the key parameters of the intertemporal collective model.

4 Empirical Implementation

The next two subsections will outline the preference and heterogeneity assumptions used in the estimation of individual preferences and the class of measurement errors that are allowed.

4.1 **Preference Assumptions**

The empirical analysis will focus on the estimation of individual preferences for private consumption and leisure. The implicit assumption is that private consumption and leisure are strongly separable from public consumption. Since this assumption is more realistic for the group of households with no children, the estimation will be performed using this restricted sample.

It is assumed that agent i's preferences can be represented using the following utility function:

$$u^{i}(c^{i}, T - h^{i}) = \frac{\left[\left(c^{i}\right)^{\sigma_{i}}\left(T - h^{i}\right)^{1 - \sigma_{i}}\right]^{1 - \rho_{i}}}{1 - \rho_{i}},$$

with $\rho_i > 0$, $0 < \sigma_i < 1$. This utility function has been used extensively in the past for its simplicity in research projects that attempt to model the relationship between consumption and leisure. Three notable examples are Kydland and Prescott, (1982) Prescott (1986), and Browning, Hansen, and Heckman (1999). The parameter ρ_i captures the intertemporal aspects of individual preferences. In particular, $-1/\rho_i$ is agent *i*'s intertemporal elasticity of substitution, which measures the willingness to substitute the composite good $(c^i)^{\sigma_i} (T - h^i)^{1-\sigma_i}$ between different dates. The parameter σ_i captures the intraperiod features of individual preferences and it measures in each period the fraction of expenditure assigned to agent *i* which is allocated to private consumption.

The consumption function $g(\bar{w}^1, h^1)$ corresponding to these preferences can be written in the following form:

$$g\left(\bar{w}^{1},h^{1}\right) = \frac{\sigma_{1}}{1-\sigma_{1}}\bar{w}_{t}^{1}\left(T-h_{t}^{1}\right)$$

4.2 Household Heterogeneity and Measurement Errors

So far the only source of household heterogeneity is the realization of the state of nature. In the estimation of individual preferences, I will allow for two additional sources of heterogeneity. A subscript h will be used to denote an observation for household h.

The main idea underlying the identification of individual preferences is that individual consumption can be written as a function of individual labor supply and own wage. In particular, given the functional form assumed for the utility functions, individual consumption and the individual value of leisure, $\bar{w}_t^1 (T - h_t^1)$, should be linearly related and therefore perfectly correlated. This implication of the model can be tested using the sample of singles, since their individual consumption is observed. In the CEX, the correlation is 0.30 for single females and 0.26 for single males. This indicates that there is a positive relationship between individual consumption and value of leisure as predicted by the model. But it also suggests that there is additional heterogeneity characterizing the function g. A potential interpretation of this finding is that, for a given $T - h^i$, the perceived value of leisure varies with age, education and seasonal dummies, because the available alternatives vary with these variables. This source of heterogeneity will be captured by assuming that agent i's utility function depends on effective leisure, \hat{l}^i , where effective leisure is defined as

$$\hat{l}_{t,h}^{i} = \left(T - h_{t,h}^{1}\right) \exp\left(\alpha_{i}^{\prime} z_{t,h}\right),\,$$

and $z_{t,h}$ is a vector containing the wife's and husband's age, an education dummy for the wife and for the husband, and a seasonal dummy. This implies that⁶

$$c_{t,h}^{1} = g\left(\bar{w}_{t,h}^{1}, h_{t,h}^{1}, z_{t,h}\right) = \frac{\sigma_{1}}{1 - \sigma_{1}} \bar{w}_{t,h}^{1} \left(T - h_{t,h}^{1}\right) \exp\left(\alpha_{i}' z_{t,h}\right).$$

As an additional source of heterogeneity, it will be assumed that the logarithm of the ratio of Pareto weights varies across households according to an unknown distribution with mean $\overline{l\mu}$. This implies that $\log(\mu_h)$ can be written in the form

$$\log\left(\mu_{h}\right) = \overline{l\mu} + \eta_{h},$$

 $^{^{6}}$ An alternative interpretation of the low correlation between consumption and value of leisure is that the assumption on preferences is restrictive.

where η is a mean-zero random variable. It is important to remark that under ex-ante efficiency the household ratio of Pareto weights cannot change over time.

Finally, to determine which class of measurement errors can be allowed in the model, I will add measurement errors in private household consumption and individual wages. It will be assumed that the measurement errors satisfy the following three conditions. First, they are additive in the logarithm of private household consumption and individual wages. Second, let $C_{t,h}^*$ and $w_{t,h}^{i*}$ be true private household consumption and wages for household h in period t and denote with $C_{t,h}$ and $w_{t,h}^i$ the observed variables. It is assumed that the true variables can be written in the following form:

$$\log C_{t,h}^* = \log C_{t,h} + \delta_C + \delta_h + \epsilon_{t,h}, \quad \log w_{ht}^{i*} = \log w_{t,h}^i + \delta_{w^i} + \delta_h + \epsilon_{t,h},$$

where δ_C and δ_w^i are two constants, δ_h is a household fixed effect, and $\epsilon_{t,h}$ is a mean-zero random variable which is common to private consumption and wages.⁷ Third, in each period t, the common component $\epsilon_{t,h}$ and the household fixed effect δ_h are independent of the information known to the household.⁸

Let $\gamma_i = \sigma_i (1 - \rho_i)$ and $\theta_i = (1 - \sigma_i) (1 - \rho_i)$. The assumptions on preferences and household heterogeneity imply that the individual Euler equations (6) can be written as follows:

$$\beta_1 E_t \left[\left(\frac{\bar{w}_{t+1,h}^1}{\bar{w}_{t,h}^1} \right)^{\gamma_1 - 1} \left(\frac{T - h_{t+1,h}^1}{T - h_{t,h}^1} \right)^{-\rho_1} e^{\alpha_1 (\gamma_1 - \rho_1 - 1) \left(z_{t+1,h} - z_{t,h} \right)} R_{t+1,h} \frac{p_t}{p_{t+1}} \right] = \frac{e^{(\gamma_1 - 1) \left(\delta_{w^i} + \delta_h + \epsilon_{t,h} \right)}}{E_t \left[e^{(\gamma_1 - 1) \left(\delta_{w^i} + \delta_h + \epsilon_{t+1,h} \right)} \right]}$$

the individual Euler equations (7) can be written in the form

$$\beta_{2}E_{t}\left[\left(\frac{C_{t+1,h}-\hat{\phi}\bar{w}_{t+1,h}^{1}\left(T-h_{t+1,h}^{1}\right)e^{\alpha_{1}z_{t+1,h}}}{C_{t}-\hat{\phi}\bar{w}_{t,h}^{1}\left(T-h_{t,h}^{1}\right)e^{\alpha_{1}z_{t,h}}}\right)^{\gamma_{2}-1}\left(\frac{T-h_{t+1,h}^{2}}{T-h_{t,h}^{2}}\right)^{\theta_{2}}\right]\\e^{\alpha_{2}\theta_{2}\left(z_{t+1,h}-z_{t,h}\right)}R_{t+1,h}\frac{p_{t}}{p_{t+1}}\right]=\frac{e^{(\gamma_{2}-1)\left(\delta_{C}+\delta_{w^{i}}+\delta_{h}+\epsilon_{t,h}\right)}}{E_{t}\left[e^{(\gamma_{2}-1)\left(\delta_{C}+\delta_{w^{i}}+\delta_{h}+\epsilon_{t+1,h}\right)}\right]},$$

and the efficiency condition (8) as

$$(\gamma_{1}-1)\log \bar{w}_{t,h}^{1} - \rho_{1}\log \left(T - h_{t,h}^{1}\right) - (\gamma_{2}-1)\log \left(C_{t,h} - \hat{\phi}\bar{w}_{t,h}^{1}\left(T - h_{t,h}^{1}\right)e^{\alpha_{1}z_{t,h}}\right) - \theta_{2}\log \left(T - h_{t,h}^{2}\right) - (\alpha_{1}\rho_{1} + \alpha_{2}\theta_{2})z_{t,h} = \overline{l\mu} + \eta_{h} + (\gamma_{1}-1)\log \hat{\phi} + (\gamma_{2}-\gamma_{1})\left(\delta_{C} + \delta_{w^{i}} + \delta_{h} + \epsilon_{t,h}\right),$$

where $\hat{\phi} = \frac{\sigma_{1}}{(1-\sigma_{1})\exp(\delta_{C})}.$

⁷Under the standard assumption that measurement errors have zero mean, the constants δ_C and δ_{w^i} must be equal to zero and the consumption and wage measurement errors must be identical.

⁸Preferences will also be estimated for single agents. In married households two respondents provide information on consumption and wages, whereas in single households only one respondent is present at the interview. To take this into account, in the estimation of preferences for singles I will also allow for measurement errors $\varepsilon_{C,t,h}^s$ and $\varepsilon_{w,t,h}^s$ that are specific to singles.

Taking the unconditional expectation of both sides, agent 1's Euler equations become

$$\beta_1 E\left[\left(\frac{\bar{w}_{t+1,h}^1}{\bar{w}_{t,h}^1}\right)^{\gamma_1 - 1} \left(\frac{T - h_{t+1,h}^1}{T - h_{t,h}^1}\right)^{-\rho_1} e^{\alpha_1(\gamma_1 - \rho_1 - 1)\left(z_{t+1,h} - z_{t,h}\right)} R_{t+1,h} \frac{p_t}{p_{t+1}}\right] = 1, \quad (10)$$

agent 2's Euler equations can be written as

$$\beta_{2}E\left[\left(\frac{C_{t+1,h}-\hat{\phi}\bar{w}_{t+1,h}^{1}\left(T-h_{t+1,h}^{1}\right)e^{\alpha_{1}z_{t+1,h}}}{C_{t,h}-\hat{\phi}\bar{w}_{t,h}^{1}\left(T-h_{t,h}^{1}\right)e^{\alpha_{1}z_{t,h}}}\right)^{\gamma_{2}-1}\left(\frac{T-h_{t+1,h}^{2}}{T-h_{t,h}^{2}}\right)^{\theta_{2}}e^{\alpha_{2}\theta_{2}\left(z_{t+1,h}-z_{t,h}\right)}R_{t+1,h}\frac{p_{t}}{p_{t+1}}\right]=1,$$
(11)

and the efficiency condition becomes

$$E\left[(\gamma_{1}-1)\log\bar{w}_{t,h}^{1}-\rho_{1}\log\left(T-h_{t,h}^{1}\right)-(\gamma_{2}-1)\log\left(C_{t,h}-\hat{\phi}\bar{w}_{t,h}^{1}\left(T-h_{t,h}^{1}\right)e^{\alpha_{1}z_{t,h}}\right)-\theta_{2}\log\left(T-h_{t,h}^{2}\right)-(\alpha_{1}\rho_{1}+\alpha_{2}\theta_{2})z_{t,h}\right]=\overline{l\mu}+(\gamma_{1}-1)\log\hat{\phi}+(\gamma_{2}-\gamma_{1})\left(\delta_{C}+\delta_{w^{i}}\right).$$
(12)

Using CEX data, the coefficients ρ_1 , γ_i , θ_i , $\hat{\phi}$, and $\overline{l\mu} + (\gamma_2 - \gamma_1) (\delta_C + \delta_{w^i})$ will be estimated by applying the Generalized Method of Moments (GMM) to these three equations. The parameters of the individual utility functions can then be recovered using the following equations:

$$\gamma_i = \sigma_i (1 - \rho_i), \quad \theta_i = (1 - \sigma_i) (1 - \rho_i), \quad \phi = \frac{\sigma_1}{1 - \sigma_1}.$$

5 Econometric Issues

The transformed Euler equations of agent 2 cannot be log-linearized. Consequently, individual preferences will be estimated using the non-linear transformed Euler equations jointly with the efficiency condition. The non-linearities in the Euler equations imply that I can only allow for the class of measurement errors introduced in the previous section, which is a special case of the measurement errors that can be allowed in linear models. To evaluate the effect of the non-linearities on the coefficient estimates, the sample of single households will be used. In particular, the estimation of individual preferences for single households requires only the transformed Euler equation of agent 1, which can be log-linearized. The individual preferences can therefore be estimated using a linear and a non-linear version of the model. The estimates can then be compared to examine the effect of a larger class of measurement errors.

The identification result of Proposition 1 holds only if the consumption function g is well defined. This requires that at least one household member decides to supply a positive amount of labor in two consecutive periods. In the sample of couples used in the estimation the fraction of households in which the husband supplies a positive amount of labor for the entire survey period is around 80%, whereas the fraction in which the wife works during the survey is around 70%. In spite of this, in the estimation the wive's labor supply and wage will be used to derive the consumption function g for the following two reasons. First, there is more variation in the labor supply of married women relative to married men. Second, the correlation between individual consumption and value of leisure is higher for single females relative to single males, which suggests that the correlation should be higher for married women relative to married men. All this implies that the sample used in the estimation can be composed only of households in which the wife works during the survey period. As a consequence, if the residuals of the transformed Euler equations are correlated with the labor force participation decisions of the wife, the estimation results will be affected by a selection bias.

To quantify the selection bias I will use the sample of single households. Denote with D_t^1 a dummy equal to 1 if agent 1 works in period t and let ζ_{t+1}^i be the error term corresponding to the transformed Euler equations of agent i. Since individual preferences are estimated using the sample of households in which agent 1 works at t and t + 1, the parameter estimates of both couples and singles are unbiased only if

$$E\left[\zeta_{t+1}^{i} \left| D_{t}^{1} = 1, D_{t+1}^{1} = 1\right] = 0,$$

i.e. only if ζ_{t+1}^i is independent of the participation decisions in period t and t+1. Suppose this independence assumption is not satisfied. For both couples and singles, the labor force participation decision of agent 1 can be formulated using the model proposed in this paper. In particular, in each period t agent 1's marginal rate of substitution between leisure and consumption is equal to the real wage if $D_t^1 = 1$, but it is greater than the real wage if $D_t^1 = 0$. Under the functional form assumptions for the individual utilities, this implies that

$$\begin{split} c_t^1 &\leq \phi \bar{w}_t^1 T \quad \text{if } D_t^1 = 1, \\ c_t^1 &> \phi \bar{w}_t^1 T \quad \text{if } D_t^1 = 0. \end{split}$$

It is assumed that the wage equation is determined outside the model and that it can be written as

$$\log w_t^1 = X_t \beta + e_t$$

where X_t includes labor market experience, its square, and a price index that is household and region specific. The selection equations in period t can then be written in the form

$$\log c_t^1 - \log \phi - \log T - X_t \beta \le e_t \quad \text{if } D_t^1 = 1,$$
$$\log c_t^1 - \log \phi - \log T - X_t \beta > e_t \quad \text{if } D_t^1 = 0.$$

Suppose that ζ_{t+1}^i , e_t , and e_{t+1} are normally distributed with mean vector 0 and covariance matrix

$$\left[\begin{array}{ccc} \sigma_{\zeta^i}^2 & \rho_{\zeta^i,t} & \rho_{\zeta^i,t+1} \\ & 1 & \rho \\ & & 1 \end{array}\right].$$

Then, by Tunali (1986),

$$E\left[\zeta_{t+1}^{i} \left| D_{t}^{1} = 1, D_{t+1}^{1} = 1\right.\right] = \sigma_{\zeta^{i}} \rho_{\zeta^{i}, t} \xi_{t}^{i} + \sigma_{\zeta^{i}} \rho_{\zeta^{i}, t+1} \xi_{t+1}^{i},$$

where

$$\xi_{t} = \frac{\phi(X_{t}\beta)\Phi\left(\frac{X_{t+1}\beta - \rho X_{t}\beta}{(1-\rho^{2})^{1/2}}\right)}{G(X_{t}\beta, X_{t+1}\beta, \rho)}, \quad \xi_{t+1} = \frac{\phi(X_{t+1}\beta)\Phi\left(\frac{X_{t}\beta - \rho X_{t+1}\beta}{(1-\rho^{2})^{1/2}}\right)}{G(X_{t}\beta, X_{t+1}\beta, \rho)},$$

and where ϕ , Φ and G are, respectively, the standard univariate normal density function, the standard univariate normal distribution function, and the standard bivariate normal distribution function. Individual consumption is observed for singles without children. Consequently, for this group of households the individual Euler equations can be estimated jointly with the labor force participation equations to quantify the selection bias. Following Newey and McFadden (1994), the Euler equations adjusted for selection are estimated using GMM in one step by adding as moment conditions the first order conditions of the bivariate probit, which determines the probability of being in one of the four possible labor supply states defined by D_t^1 and D_{t+1}^1 . A similar approach could be used for couples, but stronger assumptions are required.⁹

The residuals of the individual Euler equations contain the expectation error implicit in these intertemporal optimality conditions. Since part of the expectation error is generated by aggregate shocks, it could be correlated across households. As suggested by Chamberlain (1984), this implies that the Euler equations can be consistently estimated only if the sample period covered by the data is long enough to contain all the stages of the business cycle. For this reason, data from 1982 to 1998 are used in the estimation.

The Euler equations and the efficiency conditions will be estimated using the continuous updating GMM. The choice of this GMM estimator is based on work by Hansen, Heaton, and Yaron (1996) and Donald and Newey (2000) indicating that the continuous updating GMM estimator has smaller bias than the more common two-step efficient GMM estimator, with and without autocorrelation. Under the assumption of rational expectations, any variable known at time t should be a valid instrument for GMM. The existence of measurement errors, however, may introduce dependence between variables known at time t and concurrent and future variables, even under rational expectations. To address this problem, only variables known at t - 1 are used. This requires three consecutive observations for the same household: two to compute the growth rate for consumption,

⁹To estimate agent 1's participation equations for couples, individual consumption must be substituted out using the first order conditions for consumption, which depend on the budget constraint multiplier in the corresponding period. Using the Euler equations and an approach similar to Heckman and MaCurdy (1980) and Browning et al. (1986), the multiplier in each period can be written as a function of the multiplier at 0 and the sequence of interest rates. If a long panel is available, the participation equations can therefore be estimated jointly with the individual Euler equations and efficiency conditions by using a fixed effect estimator. Unfortunately, the panel used in this paper covers only two consecutive period, which implies that the participation equations can be estimated only if it is assumed that the initial multiplier is constant across households or by using as a proxy for the multiplier initial wealth.

leisure, and wages, and at least one additional observation to construct the instrument set. In the CEX, labor supply and labor income data are only measured in the first and last interview, which implies that only two consecutive observations are available for each household. To address this problem, the set of instruments is constructed employing lagged cohort variables, where the cohort variables are computed using 7-years intervals for the head's year of birth.

6 CEX Data

The CEX survey is a rotating panel organized by the Bureau of Labor Statistics (BLS). Each quarter about 4,500 households, representative of the U.S. population, are interviewed: 80 percent are reinterviewed the following quarter, while the remaining 20 percent are replaced by a new randomly selected group. Each household is interviewed at most for four quarters and detailed information is collected on expenditures, income, and demographics. Following Meghir and Weber (1996) household level data for the available quarters are used in the estimation. The sample employed in this paper covers the period 1982-1998. The first two years are excluded because the data were collected with a slightly different methodology.

The CEX collects consumption data in each quarter of the survey. Labor supply and labor income data, however, are gathered only during the first and last interviews unless a member of the household reports changing his or her employment. In the second and third interviews the labor variables are set equal to the data reported in the first interview. Consequently, in the estimation I use quarterly variables computed using the first and last interviews.

Quarterly household consumption of singles is computed as the sum of food at home, food away from home, tobacco, alcohol, public and private transportation, personal care, clothing, house maintenance, heating fuel, utilities, housekeeping services, and transportation repairs, which is the definition used in Attanasio and Weber (1995). Household consumption of couples is obtained by subtracting the expenditure on goods that are clearly public consumption from the definition used for singles, namely house maintenance, heating fuel, and housekeeping services. Quarterly individual labor supply is calculated as the number of hours usually worked per week multiplied by 13 weeks. The total amount of time that an agent can divide between labor supply and leisure, T, is set equal to 1183, which is equal to 13 hours per day times 7 days a week times 13 weeks a quarter.¹⁰ Quarterly leisure can then be computed as T minus quarterly labor supply. The individual hourly wage rate is determined using three variables: the amount of the last gross pay, the time period the last gross pay covered, and the number of hours usually worked per week in the corresponding period. The after-tax wage rate is computed using federal effective tax rates

¹⁰The 13 hours per day are computed by allocating 8 hours to sleep, 1 hour to the time required to reach the workplace, and 2 hours to exogenous household production. I also experimented with 12 and 14 hours per day. This change has a small effect on the estimation of σ , which can be explained by noting that, for any level of labor supply, T determines the amount of leisure. However, the main findings of the paper do not change. An alternative approach would be to use a time survey to compute T for married females and males, and for single females and males.

generated by the NBER's TAXSIM model. The gross interest rate is obtained compounding the 20-year municipal bond rate for the three quarters that separate the first interview from the last. Household consumption, individual after-tax wages, and the gross interest rate are deflated using a household specific price index. The index is calculated as a weighted average of the consumer price indices published by the Bureau of Labor Statistics, with weights equal to the expenditure share for the particular consumption good.

The identification result requires that at least one household member supplies a positive amount of labor in two consecutive periods. Consequently, I drop from the sample couples in which the wife does not work during at least one of the two quarters used in the estimation. For singles, I drop a household if the head does not work in one of the two quarters. Households with children and households in which the head is older than 65 and younger than 22 are also excluded. Households with missing values in one of the variables defining the individual Euler equations and efficiency condition are dropped. For couples, a household is not used in the estimation if the husband's or the wife's labor supply is lower than 20 hours, or the wife's real after-tax hourly wage is larger than 50 dollars. For singles, I drop a household if the head's labor supply is less than 20 hours or the real after-tax hourly wage is larger than 50 dollars.¹¹ Summary statistics in 1984 dollars for the main variables are reported in table 1.

Independent Variable	Mean for Singles	Mean for Couples
Real Consumption per Quarter	1563.8	2545.1
Head's Labor Supply per Week	42.7	44.4
Spouse's Labor Supply per Week	-	32.1
Conditional Spouse's Labor Supply per Week	-	38.8
Head's After Tax Wage per Hour	7.7	9.2
Wife's Before Tax Wage per Hour	-	6.6
Number of Observations	9464	5064
Number of Families	2366	1266

7 Results

To evaluate the performance of the identification result, individual preferences are initially estimated for single agents using several specifications. First, preferences are estimated using standard

¹¹The fraction of couples in which the wife's wage is larger than 50 dollars is around 0.5 percent. The fraction of single males and females is around 0.2 percent. The fraction of couples in which the wife works less than 20 hours is 5 percent of the sample. The fraction of singles in which the head works less than 20 hours is around 1 percent for males and around 2 percent for females.

household consumption Euler equations and the intraperiod condition. Under the assumptions on preferences and heterogeneity of section 4, the two equations can be written as follows:

$$E\left[\left(\frac{C_{t+1,h}}{C_{t,h}}\right)^{\gamma-1} \left(\frac{T-h_{t+1,h}}{T-h_{t,h}}\right)^{\theta} e^{\alpha\theta\left(z_{t+1,h}-z_{t,h}\right)} \beta R_{t+1,h} \frac{p_{t,h}}{p_{t+1,h}}\right] = 1,$$
(13)

$$E\left[\log C_{t,h} - \log \hat{\phi} - \log\left(\bar{w}_{t,h}^1 \left(T - h_{t,h}^1\right)\right) - \alpha' z_{t,h}\right] = 0.$$
(14)

This specification corresponds to the approach traditionally used by the intertemporal literature, except that the intraperiod condition is included in the estimation to pin down the intraperiod parameter σ .¹² Second, preferences of singles are estimated using agent 1's transformed consumption Euler equations (10) and the identification result. Note that the transformed Euler equations (10) contain the same information as equations (13) and (14), since they are obtained by substituting the intraperiod condition in the standard Euler equations.

The standard estimation and the estimation based on the identification result will be implemented using log-linearized as well as non-linear Euler equations. All specifications are estimated with and without selection correction terms.

The results for the log-linearized version of the model are reported in table 2 for females and table 3 for males. The estimates obtained using the identification result are similar to the ones obtained using the standard method. The estimates for the coefficient ρ are about 1.7 for single males and 5 for single females. The parameter σ is precisely estimated only if the intraperiod condition is added to the estimation as an additional moment condition. In this case the wife's σ is around 0.15, whereas the husband's is around 0.50. Note that in the estimation of couples' preferences an intraperiod condition will be used in the form of the efficiency equation. This will enable me to precisely estimate σ .

The results obtained using the non-linear version of the model are reported in table 4 for single females and table 5 for single males and are similar to the estimates obtained using the log-linearized Euler equations. This suggests that the estimation results do not vary if measurement errors and unobserved heterogeneity are generalized to the class that can be allowed in linear models. The addition of the selection terms to the model does not produce significant differences in the results, which are reported in tables 6, 7, 8, and 9. In all specifications the selection terms are never statistically significant. This finding can be interpreted in two different ways. Either I am not able to precisely estimate the labor force participation decision, or the unobservable heterogeneity in the participation decision is independent of the Euler equation error term. In most specifications both experience and its square have a statistically significant effect on the participation decision. This suggests that the second interpretation is plausible and that selection biases should not have significant effects on the estimation of individual preferences for couples.

¹²Theoretically, σ can be estimated using only the standard household consumption Euler equations. But empirically σ can be precisely estimated only if the intraperiod condition is added to the estimation.

The main empirical results are the estimates for couples, which are obtained using the transformed Euler equation (10) for the wife, the transformed Euler equation (11) for the husband, and the efficiency condition (12). The results are reported in tables 10. The wife's ρ is estimated to be around 4.4, whereas the husband's ρ is estimated to be around 2.5. Moreover, the difference between the wife's estimated ρ and the husband's is statistically significant. The wife's and husband's σ are estimated to be around 0.28. Table 10 also reports the estimate of the mean relative decision power under the assumption that the constants δ_C and δ_{w^i} in the measurement errors are equal to zero. Note that the estimation of the model produces an estimate of the logarithm of the expected value of μ . The reported estimate is obtained by taking a first order Taylor expansion of it. The mean relative decision power is measured to be around 0.82, but the standard errors are five times as large. The last column of table 10 reports the results of the estimation of the standard unitary model with separability between consumption and leisure. The estimate of ρ is 3.7, which is between the estimated ρ for married females and married males and within the range of estimates obtained in the past.

Finally, to test the assumption of ex-ante efficiency, the individual Euler equations (10) and (11) are also estimated without including the efficiency condition (12). Using a distance statistic test with one degree of freedom, ex-ante efficiency cannot be rejected at any standard significance level.

To understand which features of the data generate the differences in intertemporal elasticities of substitution between females and males, consider a single agent. The parameter σ measures the consumption budget share of this individual. In the CEX, the average consumption budget share is around 0.25 for both single females and males. These numbers are consistent with the estimates obtained in this paper which are between 0.12 and 0.55.

To determine how ρ is identified, for a given σ define the composite good $\bar{C} = c^{\sigma} l^{1-\sigma}$. Note that if a single agent decides to save one unit of \bar{C} in period t, she will be able to increase consumption at t+1 by $R_{t+1} (p_t/p_{t+1})^{\sigma} (w_t/w_{t+1})^{1-\sigma}$. We can therefore interpret $R_{t+1} (p_t/p_{t+1})^{\sigma} (w_t/w_{t+1})^{1-\sigma}$ as the gross return on \bar{C} , where R_{t+1} captures the return for investing one unit of \bar{C} in the risk-free asset, and $(p_t/p_{t+1})^{\sigma}$ and $(w_t/w_{t+1})^{1-\sigma}$ measure the change in prices between t and t+1 of the two goods that form \bar{C} , weighted using the corresponding budget shares. Using the log-linearized model, it is straightforward to show that $1/\rho$ measures the percentage change in \bar{C}_{t+1}/\bar{C}_t generated by a one percent increase in $R_{t+1} (p_t/p_{t+1})^{\sigma} (w_t/w_{t+1})^{1-\sigma}$. This elasticity can be determined in the CEX by implementing an IV regression of the logarithm of \bar{C}_{t+1}/\bar{C}_t on the logarithm of $R_{t+1} (p_t/p_{t+1})^{\sigma} (w_t/w_{t+1})^{1-\sigma}$. If σ is set equal to the average consumption budget share, the estimated coefficient is 0.16 for single females and 0.56 for single males, which explains the estimated heterogeneity in intertemporal preferences.¹³

¹³I use an IV regression instead of an OLS regression to replicate the GMM estimation and to take into consideration that labor supply is used to construct the dependent variable \bar{C} as well as the regressor $R_{t+1} (p_{t+1}/p_t)^{\sigma} (w_{t+1}/w_t)^{1-\sigma}$.

It is now straightforward to understand how the data generates a different ρ for males and females. Consider two identical single households except that the first one has a female head whereas the second one has a male head. Since males and females have identical σ , these two households face the same return on \bar{C} and given this return they choose \bar{C}_{t+1}/\bar{C}_t optimally. Consider an increase in the rate of return. In the data both single females and males increase the ratio \bar{C}_{t+1}/\bar{C}_t . However, the increase in period t+1 consumption relative to period t is larger for males, which suggests that females have a higher willingness to pay for a smooth consumption path.

Two remarks are in order. First, there is weak evidence of selection in the marriage market based on individual preferences. In particular, agents that are at the extremes of the risk aversion distribution are less likely to be married. Second, according to the results, the elasticity of intertemporal substitution for males is around twice the elasticity for females. Since in the proposed model the parameter ρ is the coefficient of relative risk aversion for the composite good $(c^i)^{\sigma_i} (T - h^i)^{1-\sigma_i}$, the results also imply that females are more risk averse than males. In Mazzocco (forthcoming) it is shown that the standard unitary model represents a good approximation of household intertemporal behavior if and only if the individual preferences satisfy a generalization of Gorman aggregation to an intertemporal framework. The heterogeneity in the estimated ρ implies that Gorman aggregation is not satisfied. Consequently, simulations of competing policies based on the standard unitary model are generally misleading, because they do not consider the full extent of intrahousehold risk sharing and specialization that can be obtained if individual preferences are heterogeneous.

One example is the evaluation of the adequacy of household saving at the time of retirement. As shown in Mazzocco (2004), the effect of risk sharing on household saving can be divided into two parts. First, individual members pool their earnings and consequently eliminate part of the uncertainty faced by the household. Under convex marginal utilities, income pooling always has the intuitive effect of reducing saving. Second, household members insure each other by allocating pooled income according to individual risk preferences and decision power. This insurance component of risk sharing can have the counterintuitive effect of raising household saving. The heterogeneity in risk aversion reported in this paper indicates that the insurance component explains a significant fraction of the accumulation and reduction of household wealth. However, as shown in Mazzocco (2004), the unitary model, and therefore any simulation based on it, completely ignores this component of risk sharing. The traditional justification for using the unitary model in simulations in spite of this drawback is that there are no estimates that can be used to fix the parameters that characterize the individual intertemporal preferences. The estimates provided in this paper fill this void.

8 Conclusions

In this paper it is shown that the preferences of each decision maker in the household can be identified and estimated even if individual consumption is not observed. The main finding is that there is a significant difference in individual preferences, with the wife exhibiting a greater desire for smooth consumption.

The main implication of this result is that intertemporal decisions cannot be analyzed using a unique utility function for the entire household, because this approach ignores important aspects of intra-household risk sharing and specialization. This implies that any policy analysis related to household intertemporal decisions should be implemented by characterizing each household member by means of individual preferences.

The analysis can be extended in at least one directions. In this paper it is assumed that the time devoted to household production is exogenously given. Under this assumption, it can be incorporated in the available time T. An important project which is left for future research is to generalize the identification result to an environment that allows for endogenous choices of domestic labor. In the meanwhile, empirical works should model T as a function of exogenous variables that determine domestic labor. In this way, differences across households in domestic labor are captured by the heterogeneity in T.

References

- Attanasio, Orazio P. and Guglielmo Weber: "Is Consumption Growth Consistent with Intertemporal Optimization? Evidence from the Consumer Expenditure Survey," *Journal of Political Economy*, Vol. 103, No. 6. (1995), pp. 1121-1157.
- [2] Blundell, R., P. A. Chiappori, T. Magnac, and C. Meghir: "Collective Labor Supply: Heterogeneity and Nonpartecipation," Institute for Fiscal Studies Working Paper (2001).
- [3] Browning, Martin: "Modelling Commodity Demands and Labour Supply with M-demands," University of Copenhagen, Institute of Economics, Discussion Paper, (1998).
- [4] Browning, Martin, Francois Bourguignon, Pierre-André Chiappori, Valerie Lechene: "Income and Outcomes: A Structural Model of Intrahousehold Allocation," *Journal of Political Econ*omy, Vol. 102, No. 6. (Dec., 1994), pp. 1067-1096.
- [5] Browning, Martin and Pierre-André Chiappori: "Efficient Intra-household Allocations: A General Characterization and Empirical Tests," *Econometrica*, Vol. 66, No. 6. (Nov., 1998), pp. 1241-1278.
- [6] Browning, Martin, Angus Deaton, and Margaret Irish: "A Profitable Approach to Labor Supply and Commodity Demands over the Life-Cycle," *Econometrica*, Vol. 53, No. 3. (May, 1985), pp. 503-544.
- [7] Browning, Martin, Lars P. Hansen and James J. Heckman: "Micro Data and General Equilibrium Models," Handbook of Macroeconomics, edited by John Taylor and Michael Woodford, (1999), North Holland.
- [8] Carneiro, Pedro and James J. Heckman: "The Evidence on Credit Constraints in Post-Secondary Schooling," The Economic Journal, Vol. 112, No. 482. (Oct., 2002), pp. 705-734.
- [9] Chiappori, Pierre-André: "Rational Household Labor Supply," *Econometrica*, Vol. 56, No. 1. (Jan., 1988), pp. 63-90.
- [10] Chiappori, Pierre-André: "Collective Labor Supply and Welfare," Journal of Political Economy, Vol. 100, No. 3. (Jun., 1992), pp. 437-467.
- [11] Chiappori, Pierre-André, Richard Blundell, and Costas Meghir: "Collective Labor Supply With Children," Manuscript, Department of Economics, University of Chicago, (2004).
- [12] Donni, Olivier: "A Collective Model of Household Behavior with Private and Public Goods: Theory and Some Evidence from U.S. Data," DELTA Working Papers, (2004).
- [13] Donald, Stephen G. and Whitney K. Newey: "A Jacknife Interpretation of the Continuous Updating Estimator," *Economics Letters*, Vol. 67, No. 3. (Jun., 2000), pp. 239-243.
- [14] Fong, Yuk-fai and Junsen Zhang: "The Identification of Unobservable Independent and Spousal Leisure," *Journal of Political Economy*, Vol. 109, No. 1. (Dec., 2001), pp. 191-202.
- [15] Gross, David B. and Nicholas S. Souleles: "Do Liquidity Constraints and Interest Rates Matter for Consumer Behavior? Evidence from Credit Card Data," *Quarterly Journal of Economics*, Vol. 117, No. 1. (Feb., 2002), pp. 149-185.

- [16] Hansen, Lars Peter, John Eaton, and Amir Yaron: "Finite-Sample Properties of Some Alternative GMM Estimators," *Journal of Business and Economic Statistics*, Vol. 14, No. 3. (Jul., 1996), pp. 262-280.
- [17] Heckman, J. James and Thomas E. MaCurdy: "A Life Cycle Model of Female Labour Supply," *The Review of Economic Studies*, Vol. 47, No. 1, Econometrics Issue. (Jan., 1980), pp. 47-74.
- [18] Kydland, Finn E. and Edward C. Prescott: "Time to Build and Aggregate Fluctuations," *Econometrica*, Vol. 50, No. 6. (Nov., 1982), pp. 1345-1370.
- [19] Manser, Marilyn and Murray Brown: "Marriage and Household Decision-Making: A Bargaining Analysis," *International Economic Review*, Vol. 21, No. 1. (Feb., 1980), pp. 31-44.
- [20] Mazzocco, Maurizio: "Intertemporal Behavior and Household Structure," Manuscript, Department of Economics, University of Wisconsin (2005).
- [21] Mazzocco, Maurizio: "Savings, Risk Sharing and Preferences for Risk," American Economic Review, Vol. 94, No. 4. (Sep. 2004b), pp. 1169-1182.
- [22] Mazzocco, Maurizio: "Household Intertemporal Behavior: a Collective Characterization and a Test of Commitment," *Review of Economic Studies*, (forthcoming).
- [23] McElroy, Marjorie B. and Mary Jean Horney: "Nash-Bargained Household Decisions: Toward a Generalization of the Theory of Demand," *International Economic Review*, Vol. 22, No. 2. (Jun., 1981), pp. 333-349.
- [24] Meghir, Costas and Guglielmo Weber: "Intertemporal Nonseparability or Borrowing Restrictions? A Disaggregate Analysis using a U.S. Consumption Panel," *Econometrica*, Vol. 64, No. 5. (Sep., 1996), pp. 1151-1181.
- [25] Prescott, Edward C.: "Theory Ahead of Business Cycle Measurement," in Real Business Cycles, Real Exchange Rates and Actual Policies, *Carnegie-Rochester Conference Series on Public Policy*, edited by K. Brunner and A. H. Meltzer, 25 (Autumn, 1986), pp. 11-44.
- [26] Runkle, David E.: "Liquidity Constraints and the Permanent-income Hypothesis: Evidence from Panel Data," *Journal of Monetary Economics*, Vol. 27, No. 1. (Feb., 1991), pp. 73-98.
- [27] Schultz, Paul T.: "Testing the Neoclassical Model of Family Labor Supply and Fertility," The Journal of Human Resources, Vol. 25, No. 4. (Autumn, 1990), pp. 599-634.
- [28] Thomas, Duncan: "Intra-Household Resource Allocation: An Inferential Approach," Journal of Human Resources, Vol. 25, No. 4. (Autumn, 1990), pp. 635-664.
- [29] Zeldes, Stephen P.: "Consumption and Liquidity Constraints: an Empirical Investigation," *Journal of Political Economy*, Vol. 97, No. 2. (Apr., 1989), pp. 305-346.

A Proofs

A.1 Lemma 1

The following Lemma determines the condition under which the marginal rate of substitution function q can be inverted and therefore the consumption function g is well-defined.

Lemma 1 The function $g(\bar{w}_t^1, h_t^1)$ is well-defined if

$$u_{lc}^{1}\left(c_{t}^{1}, T-h_{t}^{1}\right)u_{c}^{1}\left(c_{t}^{1}, T-h_{t}^{1}\right)-u_{cc}^{1}\left(c_{t}^{1}, T-h_{t}^{1}\right)u_{l}^{1}\left(c_{t}^{1}, T-h_{t}^{1}\right)\neq0,$$
(15)

for any realization of the exogenous variables.

Proof. For any realization of the exogenous variables define

$$d^{1}\left(c^{1},h^{1},\bar{w}^{1}\right) = q^{1}\left(c^{1}_{t},h^{1}_{t}\right) - \bar{w}^{1}_{t} = 0.$$

By the implicit function theorem, $g^1(\bar{w}^1, h^1)$ is well-defined if $\frac{\partial d^1}{\partial c^1} \neq 0$. Which implies condition (15).

A.2 Proof of Propositions 1

Given the assumption that preferences are separable over time and across states of nature, the household decision process can be divided into two stages. In the first stage, resources are optimally allocated to each period and state of nature. Let $Y_{t,\omega}$ the optimal amount of resources allocated to period t and state ω . In the second stage, the household chooses optimal consumption and leisure in each period and state of nature given $w_{t,\omega}^1$, $w_{t,\omega}^2$, $p_{t,\omega}$, and $Y_{t,\omega}$ according to the following problem:

$$\max_{\substack{c_{t,\omega}^1, c_{t,\omega}^2, l_{t,\omega}^1, l_{t,\omega}^2, l_{t,\omega}^2 \\ s.t. \sum_{i=1}^2 p_{t,\omega} c_{t,\omega}^i \leq \sum_{i=1}^2 w_{t,\omega}^i h_{t,\omega}^i + Y_{t,\omega}}$$

The price of the private good, $p_{t,\omega}$, agent 1's wage, $w_{t,\omega}^1$, and agent 2's wage, $w_{t,\omega}^2$, represent four independent sources of exogenous variation. The fifth source of variation is $Y_{t,\omega}$. It is important to remark that $Y_{t,\omega}$ is endogenously determined and it is a function of the exogenous variables in any period and state of nature. This has two implications. First, a change in one of the exogenous variables at $t' \neq t$ and $\omega' \neq \omega$ varies $Y_{t,\omega}$. Second, a change in an exogenous variable at $t' \neq t$ and $\omega' \neq \omega$ can vary household decisions in period t and state ω only through $Y_{t,\omega}$. In the remainder of the proof a change in $Y_{t,\omega}$ should be interpreted as a change in an exogenous variable in period t' and state ω' that varies $Y_{t,\omega}$.

Consider first the case in which both agents work. Note that if the function $g\left(\bar{w}_t^1, h_t^1\right)$ can be identified, the original marginal utilities can also be identified by means of the transformed marginal utilities which are known. In the remainder of the proof it will be shown that $\frac{\partial g}{\partial \bar{w}^1}$ and $\frac{\partial g}{\partial h^1}$ can be identified, which implies that $g\left(\bar{w}^1, h^1\right)$ can be identified up to an additive constant. Consider an arbitrary period t and state ω . Given w^1 , w^2 , p, and Y, optimal household private consumption, agent 1's labor supply, and agent 2's labor supply can be written in the following form:

$$C = C(w^{1}, w^{2}, p, Y), h^{1} = h^{1}(w^{1}, w^{2}, p, Y), h^{2} = h^{2}(w^{1}, w^{2}, p, Y).$$

Agent 2's transformed marginal utility of private consumption is defined as follows:

$$f_c^2(C,\bar{w}^1,h^1,h^2) = u_c^2(C - g(\bar{w}^1,h^1),T - h^2).$$
(16)

By construction this equation is satisfied for any combination of w^1 , w^2 , p, and Y. Consider an arbitrary w^1 , w^2 , p, and Y. Let dw^1 , dw^2 , dp, and dY be a small change in the exogenous variables with the following properties: (i) $dw_1 = \frac{w_1}{p}dp$, which implies that $d\bar{w}_1 = 0$; (ii) dw_2 , dp, and $d\bar{Y}$ are the solution of the following linear system:

$$\frac{\partial C}{\partial w_2} dw_2 + \left(\frac{\partial C}{\partial w_1} \frac{w_1}{p} + \frac{\partial C}{\partial p}\right) dp + \frac{\partial C}{\partial \bar{Y}} d\bar{Y} = dC \neq 0,$$

$$\frac{\partial h^1}{\partial w_2} dw_2 + \left(\frac{\partial h^1}{\partial w_1} \frac{w_1}{p} + \frac{\partial h^1}{\partial p}\right) dp + \frac{\partial h^1}{\partial \bar{Y}} d\bar{Y} = dh^1 = 0,$$

$$\frac{\partial h^2}{\partial w_2} dw_2 + \left(\frac{\partial h^2}{\partial w_1} \frac{w_1}{p} + \frac{\partial h^2}{\partial p}\right) dp + \frac{\partial h^2}{\partial \bar{Y}} d\bar{Y} = dh^2 = 0,$$

i.e., the change varies household private consumption, but agent 1's labor supply and agent 2's labor supply stay constant. The change in f_c^2 implied by dw_1 , dw_2 , dp, and $d\bar{Y}$ can be computed as follows:¹⁴

$$df_c^2 = \frac{\partial f_c^2}{\partial C} dC + \frac{\partial f_c^2}{\partial \bar{w}_1} d\bar{w}_1 + \frac{\partial f_c^2}{\partial h^1} dh^1 + \frac{\partial f_c^2}{\partial h^2} dh^2 = \frac{\partial f_c^2}{\partial C} dC$$

Similarly, the change in u_c^2 implied by dw_1 , dw_2 , dp, and $d\bar{Y}$ can be written in the following form:

$$du_c^2 = -\frac{\partial u_c^2}{\partial c^2} dC$$

Since equation (16) is satisfied for any w_1 , w_2 , p, and \overline{Y} , the change in f_c^2 must equal the change in u_c^2 . Consequently,

$$\frac{\partial f_c^2}{\partial C} = -\frac{\partial u_c^2}{\partial c^2}.$$
(17)

Since $\frac{\partial f_c^2}{\partial C}$ is known, $\frac{\partial u_c^2}{\partial c^2}$ is also known.

Consider a change dw_1 , dw_2 , dp, and $d\bar{Y}$ with the following properties: (i) $dw_1 = \frac{w_1}{p}dp$, which implies that $d\bar{w}_1 = 0$; (ii) dw_2 , dp, and $d\bar{Y}$ are the solution of the following linear system:

$$\frac{\partial C}{\partial w_2} dw_2 + \left(\frac{\partial C}{\partial w_1} \frac{w_1}{p} + \frac{\partial C}{\partial p}\right) dp + \frac{\partial C}{\partial \bar{Y}} d\bar{Y} = dC = 0,$$

$$\frac{\partial h^1}{\partial w_2} dw_2 + \left(\frac{\partial h^1}{\partial w_1} \frac{w_1}{p} + \frac{\partial h^1}{\partial p}\right) dp + \frac{\partial h^1}{\partial \bar{Y}} d\bar{Y} = dh^1 \neq 0,$$

$$\frac{\partial h^2}{\partial w_2} dw_2 + \left(\frac{\partial h^2}{\partial w_1} \frac{w_1}{p} + \frac{\partial h^2}{\partial p}\right) dp + \frac{\partial h^2}{\partial \bar{Y}} d\bar{Y} = dh^2 = 0,$$

¹⁴Alternatively, one could totally differentiate f_c^2 with respect to the exogenous variables w_1 , w_2 , p, and \bar{Y} and then impose the constraints implied by the system of linear equations.

i.e., the change varies agent 1's labor supply, but household private consumption and agent 2's labor supply stay constant. According to equation (16), the implied change in f_c^2 must equal the implied change in u_c^2 . Consequently, the following equation must be satisfied:¹⁵

$$\frac{\partial f_c^2}{\partial h^1} = -\frac{\partial u_c^2}{\partial c^2} \frac{\partial g}{\partial h^1}$$

Since $\frac{\partial f_l^2}{\partial h^1}$ and $\frac{\partial u_c^2}{\partial c^2}$ are known, $\frac{\partial g}{\partial h^1}$ is identified.

Consider a change dw_1 , dw_2 , dp, and $d\bar{Y}$ with the following properties: (i) $dw_1 \neq \frac{w_1}{p}dp$, which implies that $d\bar{w}_1 \neq 0$; (ii) dw_1 , dw_2 , dp, and $d\bar{Y}$ are the solution of the following linear system:

$$\frac{\partial C}{\partial w_1} dw_1 + \frac{\partial C}{\partial w_2} dw_2 + \frac{\partial C}{\partial p} dp + \frac{\partial C}{\partial \bar{Y}} d\bar{Y} = dC = 0,$$

$$\frac{\partial h^1}{\partial w_1} dw_1 + \frac{\partial h^1}{\partial w_2} dw_2 + \frac{\partial h^1}{\partial p} dp + \frac{\partial h^1}{\partial \bar{Y}} d\bar{Y} = dh^1 = 0,$$

$$\frac{\partial h^2}{\partial w_1} dw_1 + \frac{\partial h^2}{\partial w_2} dw_2 + \frac{\partial h^2}{\partial p} dp + \frac{\partial h^2}{\partial \bar{Y}} d\bar{Y} = dh^2 = 0,$$

i.e., the change does not vary household private consumption, agent 1's labor supply, and agent 2's labor supply. By equation (16), the implied change in f_c^2 must equal the implied change in u_c^2 , which implies that the following equation must be satisfied:

$$\frac{\partial f_c^2}{\partial \bar{w}_1} = -\frac{\partial u_c^2}{\partial c^2} \frac{\partial g}{\partial \bar{w}_1}.$$

Since $\frac{\partial f_c^2}{\partial \bar{w}_1}$ and $\frac{\partial u_c^2}{\partial c^2}$ are known, $\frac{\partial g}{\partial \bar{w}_1}$ is identified. Since $\frac{\partial g}{\partial h^1}$ and $\frac{\partial g}{\partial \bar{w}_1}$ are known, the function g is identified up to the constant of integration. It is then straightforward to use $g(\bar{w}_1, h^1)$ to recover u_c^i, u_l^i , and from f_c^i and f_l^i up to the additive constant of q.

It is important to remark that the proof requires that the following matrix of coefficients of the linear systems:

$$\begin{bmatrix} \frac{\partial C}{\partial w_2} & \frac{\partial C}{\partial p} & \frac{\partial C}{\partial \bar{Y}} \\ \frac{\partial h^1}{\partial w_2} & \frac{\partial h^1}{\partial p} & \frac{\partial h^1}{\partial \bar{Y}} \\ \frac{\partial h^2}{\partial w_2} & \frac{\partial h^2}{\partial p} & \frac{\partial h^2}{\partial \bar{Y}} \end{bmatrix}$$

is of full rank. There are two cases in which this condition is not satisfied: (i) at least one of the demand functions is independent of all the exogenous variables; (ii) the rows or columns are linearly dependent. Since the first case is not realistic, only the second one will be discussed. The rows of the matrix are linearly dependent if the variation in one of the demand functions generated by changes in the exogenous variables provides no additional information conditional on the variation in the other demand functions. The columns are linearly dependent if a change in one of the exogenous variables provide no additional information on how the demand functions C, h^1 , and h^2 vary conditional on the variation generated by the other exogenous variables. This emphasizes

¹⁵The steps used to derive this equation are equivalent to the steps used to derive (17).

that the identification of individual preferences requires that independent variations in C, h^1 , and h^2 are observed and that the exogenous variables can generate it.

Consider the case in which only agent 1 supplies a positive amount of labor. In this case, h^2 is always equal to zero and all the reduced-form marginal utilities are known except f_l^2 . In the first part of the proof, the equation defining f_l^2 and variation in h^2 were never used. Consequently, the previous argument can also be applied to households in which only one agent supplies a positive amount of labor by dropping h^2 from equation (16) and the corresponding linear equation from the three linear systems. $g(\bar{w}_1, h^1)$ is therefore identified up to the additive constant and all marginal utilities are identified except u_l^2 .

B Tables

B.1 Log-linearized Euler Equations for Singles.

Table 2: Log-linearized Individual Euler Equations for Single Females: Identification Result vs Standard Methods.

Parameters	Identification	Standard
$\overline{ ho}$	5.02	5.21
	[2.49]	[1.08]
σ	0.30	0.12
	[0.44]	[0.07]
J-Statistics	15.2	37.9
$P > \chi^2$	0.71	0.73
number of observations		1228

Asymptotic standard errors in brackets. All models are estimated with GMM using the following instruments: first to second lags of after tax real wage growth, marginal tax growth; first to fourth lags of real consumption growth, income growth, gross pay growth, labor supply growth, the household specific price index growth. All instruments are calculated at the cohort level.

Parameters	Identification	Standard
$\overline{\rho}$	1.72	1.65
	[0.96]	[0.43]
σ	0.08	0.55
	[0.73]	[0.26]
J-Statistics	9.5	33.8
$P > \chi^2$	0.96	0.87
number of observations		1138

Table 3: Log-linearized Individual Euler Equations for Single Males: Identification Result vs Standard Methods.

Asymptotic standard errors in brackets. All models are estimated with GMM using the following instruments: first to second lags of after tax real wage growth, marginal tax growth; first to fourth lags of real consumption growth, income growth, gross pay growth, labor supply growth, the household specific price index growth. All instruments are calculated at the cohort level.

B.2 Non-linear Euler Equations for Singles.

Table 4: Individual Euler Equations for Single Females: Identification Result vs Standard Methods.

Parameters	Identification	Standard
$\overline{\rho}$	5.26	5.44
	[0.57]	[0.72]
σ	0.08	0.31
	[0.12]	[0.06]
J-Statistics	14.6	34.7
$P > \chi^2$	0.33	0.34
number of observations		1228

Asymptotic standard errors in brackets. All models are estimated with GMM using the following instruments: first lag of after tax real wage growth; first to third lags of real consumption growth, marginal tax growth, real gross interest rate growth; first to fourth lags of income growth, the household specific price index growth; all instruments are calculated at the cohort level.

Parameters	Identification	Standard
$\overline{ ho}$	1.96	1.62
	[0.63]	[0.39]
σ	0.45	0.51
	[0.56]	[0.24]
J-Statistics	5.5	20.9
$P > \chi^2$	0.85	0.75
number of observations		1138

Table 5: Individual Euler Equations for Single Males: Identification Result vs Standard Methods.

Asymptotic standard errors in brackets. All models are estimated with GMM using the following instruments: first and second lags of labor supply growth; first to third lags of leisure growth; first to fourth lags of income growth, log of real gross rate of return; third and fourth lags of real consumption growth; all instruments are calculated at the cohort level.

B.3 Log-linearized Euler Equations for Singles, Controlling for Selection.

Table 6:	Log-linearized	Individual	Euler	Equations	for	Single	Females:	Identification	Result	vs
Standard	Methods Cont	rolling for S	Selectio	on.						

Parameters	Identification	Standard
$\overline{ ho}$	5.40	5.03
	[2.27]	[1.23]
σ	0.16	0.10
	[0.15]	[0.05]
Inverse Mills' Ratio t	2.12	4.24
	[2.10]	[3.45]
Inverse Mills' Ratio $t + 1$	-1.69	-4.47
	[5.22]	[3.93]
J-Statistics	14.22	33.5
$P > \chi^2$	0.58	0.79
number of observations		1228

See note table 2.

Table 7: Log-linearized Individual Euler Equations for Single Males: Identification Result vs Standard Methods Controlling for Selection.

Parameters	Identification	Standard
$\overline{\rho}$	2.14	1.70
	[1.09]	[0.31]
σ	0.10	0.18
	[0.20]	[0.09]
Inverse Mills' Ratio t	0.19	-0.53
	[0.75]	[0.88]
Inverse Mills' Ratio $t + 1$	-0.65	4.62
	[2.14]	[3.76]
J-Statistics	9.3	33.2
$P > \chi^2$	0.90	0.80
number of observations		1138

See note table 3.

B.4 Non-linear Euler Equations for Singles, Controlling for Selection.

Table 8:	Non-linear	Individual	Euler	Equations	for	Single	Females:	Identification	Result	\mathbf{vs}	Stan-
dard Met	thods Cont	rolling for S	Selectio	on.							

Parameters	Identification	Standard
$\overline{ ho}$	5.41	5.13
	[0.85]	[1.11]
σ	0.14	0.32
	[0.19]	[0.08]
Inverse Mills' Ratio t	-0.74	-0.50
	[4.09]	[2.34]
Inverse Mills' Ratio $t + 1$	1.29	2.33
	[6.38]	[4.36]
J-Statistics	14.9	32.6
$P > \chi^2$	0.14	0.29
number of observations		1228

See note table 4.

Table 9: Non-linear Individual Euler Equations for Single Males: Identification Result vs Standard Methods Controlling for Selection.

Parameters	Identification	Standard
$\overline{ ho}$	1.90	1.84
	[0.65]	[0.29]
σ	0.45	0.26
	[0.35]	[0.15]
Inverse Mills' Ratio t	-0.40	2.08
	[2.63]	[2.65]
Inverse Mills' Ratio $t + 1$	0.47	-1.84
	[1.76]	[2.10]
J-Statistics	5.62	23.4
$P > \chi^2$	0.59	0.44
number of observations		1138

See note table 5.

B.5 Euler Equations for Couples.

Table 10: Individual Euler Equations for Couples with the Emiliency Condition.				
Parameters	Wife	Husband	Parameter	Unitary Model
			Difference	with Separability
$\overline{ ho}$	4.42	2.51	1.91	3.69
	[0.43]	[0.78]	[0.88]	[0.40]
σ	0.29	0.27	0.02	-
	[0.12]	[0.13]	[0.19]	
μ	0.82	-	-	-
	[4.32]			
J-Statistics			56.9	24.4
$P > \chi^2$			0.33	0.55
number of observations				1266

Table 10: Individual Euler Equations for Couples with the Efficiency Condition.

Asymptotic standard errors in brackets. The estimate of μ is obtained by computing a first order Taylor expansion under the assumption that the constants in the measurements errors are equal to zero. All models are estimated with GMM using the following instruments: first to second lags of marginal tax growth; first to third lags of wife's and husband's gross pay growth; first to fourth lags of real household consumption growth, household income growth, wife's and husband's after tax real wage growth, wife's and husband's labor supply growth.

	1	1	<i>.</i>
Parameters	Wife	Husband	Parameter Difference
$\overline{\rho}$	4.23	2.62	1.61
	[0.43]	[0.66]	[0.78]
σ	0.24	0.17	0.07
	[0.13]	[0.13]	[0.18]
μ	-	-	-
J-Statistics			56.1
$P > \chi^2$			0.36
Efficiency Test			
Distance Statistics			0.8
$P > \chi^2$			0.37
number of observations			1266

Table 11: Individual Euler Equations for Couples without the Efficiency Condition.

See note table 10.